Author’s response to reviews

Title: Sneeze and pop: a ruptured varicocele Analysis of literature, guided by a well-documented case-report

Authors:

Daan Reesink (d.reesink@antoniusziekenhuis.nl)

Peter Huisman (phuisman@tergooi.nl)

Judit Wiltink (jwiltink@tergooi.nl)

Arto Boeken Kruger (ABoekenKruger@tergooi.nl)

Tycho Lock (M.T.W.T.Lock@umcutrecht.nl)

Version: 1 Date: 11 Jul 2018

Author’s response to reviews:

Dear Jing Zhang,

Thank you for considering our manuscript for review. Hereby our reply to the revisions.

1. Thank you for providing a Consent to publish statement, however please specify if written informed consent to publish was obtained from the participant in this section. And if so, please supply the consent statement in the following form: “Written informed consent was obtained from the patient for publication of this Case Report and any accompanying images. A copy of the written consent is available for review by the Editor of this journal.” Please also provide a signed consent form by patient as a separate file when submitting your revision.

   This section has been changed in the manuscript, and a signed consent form by the patient has been provided.

2. Please amend the structure of the Abstract to ensure it includes: Background, Case presentation, Conclusions—3 subsections. Please also amend the structure of the main text to ensure it includes: Background, Case presentation, Discussion and Conclusions—3 subsections.
The abstract and manuscript have been revised to amend to the structure stated above. The text of the manuscript has been restructured too, to a more logical structure with only 3 subsections.

3. On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

Please see uploaded documents.

We look forward to your reply.

Kind regards,

Daan J Reesink